Author's response to reviews

Title: The VicGeneration study - A birth cohort to examine the environmental, behavioural and biological predictors of early childhood caries: Background, aims and methods

Authors:

Andrea M de Silva-Sanigorski (andreams@unimelb.edu.au)
Hanny Calache (calacheh@dhsy.org.au)
Mark Gussy (M.Gussy@latrobe.edu.au)
Stuart G Daspher (stuartgd@unimelb.edu.au)
Jane Gibson (jgib@unimelb.edu.au)
Elizabeth Waters (ewaters@unimelb.edu.au)

Version: 2 Date: 13 February 2010

Author's response to reviews:

All of the revisions requested have been made to the manuscript.

regards

Andrea